Author’s response to reviews

Title: The relationship between healthy behaviors and health outcomes among older adults in Russia

Authors:

Anna Selivanova (ankaselivanka@gmail.com)
Jane M Cramm (cramm@bmg.eur.nl)

Version: 3 Date: 7 September 2014

Author’s response to reviews:

Dear Dr. Cramm:

Thank you for submitting your revised manuscript.

I am sorry to call on your time again, but before we proceed further, the Editor would like to request additional modification for your manuscript.

1. Ethics Statement

We added the following text to the manuscript:

ACKNOWLEDGEMENTS

This work was supported by a grant (Grant number: 60-61900-98-130) from the Netherlands Organization for Health Research and Development (ZonMw) as part of the National Care for the Elderly Programme. The study complies with the rules as stated in the Declaration of Helsinki. The medical ethics committee of the Erasmus Medical Centre, Rotterdam, the Netherlands, approved methods for data collection (protocol number MEC2011-041).